Author’s response to reviews

Title: Use of Real time Polymerase Chain Reaction for detection of M. tuberculosis, M. avium and M. kansasii from clinical specimen

Authors:

Arnold Bainomugisa (arnoldbaino@yahoo.co.uk)
Chris Muchwa (muchwachris@yahoo.com)
Joseph Akol (jpal01@gmail.com)
Paul Mubiri (mubirip@gmail.com)
Sam Ogwang (sxo81@case.edu)
Enock Matovu (matovue@vetmed.mak.ac.ug)
Eddie Wampande (wamps@vetmed.mak.ac.ug)
Moses Joloba (mlj10@case.edu)

Version: 8
Date: 28 March 2015

Author’s response to reviews: see over
Authors’ response to reviews

Title: Use of Real Time Polymerase Chain Reaction for Detection of M. tuberculosis, M. avium and M. kansasii From Clinical Specimens

Authors:

* Bainomugisa Arnold²,³ (arnoldbaino@yahoo.co.uk), Muchwa Chris²,³ (muchwachris@yahoo.com), Akol Joseph²,³ (jpakol01@gmail.com), Mubiri Paul³ (mubirip@gmail.com), Ogwang Sam²,³ (sxo81@case.edu), Matovu Enock ⁴ (matovue@vetmed.mak.ac.ug), Wampande Eddie¹,⁴ (wamps@vetmed.mak.ac.ug), Joloba Moses¹,³ (mlj10@case.edu)

*Corresponding author:

Bainomugisa Arnold, Joint Clinical Research Centre, P.O Box 10005, Lubowa-Kampala

Version: 5

Date: 28/03/2015

Authors response: Appear in red below
Editors’ comments:

1. Re-name Abstract Introduction to Background
   Changed the Abstract introduction to Background

2. Include the Methods section between the Background and Results sections
   Transferred methods section to appear after Background

3. List the figure legends after the References
   Listed the figure legends after the references